Author’s response to reviews

Title: Determinants of Severe Acute Malnutrition Among Under Five Children In rural Enebsie Sarmidr District, East Gojjam Zone, North West Ethiopia, 2016

Authors:

Abate Awoke (abateawoke@yahoo.com)
Mulatu Ayana (mulatuayana@gmail.com)
Tenaw Gualu (tenawgualu@yahoo.com)

Version: 4 Date: 15 Aug 2017

Author’s response to reviews:

In all the sections of the manuscript (introduction, methodology, result and conclusion), I have tried to make the manuscript scientific.

But the methodology part remains in detail as per the comment given by the reviewers during the first review process. The whole of the manuscript was edited by two people fluent in English.